Author's response to reviews

Title: Maternal morbidity in the first year after childbirth in Mombasa Kenya; a needs assessment

Authors:

Matthew F Chersich (chersich@doctors.org.uk)
Nicole Kley (nckley@yahoo.de)
Stanley MF Luchters (stanley.luchters@ugent.be)
Carol Njeru (Carol.Njeru@icrhk.org)
Elodie Yard (eloyard@yahoo.fr)
Mary J Othigo (mj_othigo@yahoo.com.au)
Marleen Temmerman (marleen.temmerman@ugent.be)

Version: 3 Date: 26 October 2009

Author's response to reviews:

Dear madam/sir,

Many thanks for your assistance with this process and the clear instructions. We have formatted the manuscript accordingly, please see full comments below.

Formatting changes requested

-----------------------------

Major revisions

Table vertical elements - Please note that we are unable to display vertical lines within the tables nor are we able to correctly display merged cells where the merged cell crosses rows - please re-layout your table without these. We are unable to write text vertically within the cells

We removed the column which was making the table have merged cells and the vertical text was removed.

Reference 1, 4, 7, 27, 29 - Please provide details of the publishing body and the place of publication

All these reference were actually electronic resources, we added the Link / URL to the references.

Reference 13, 49 - please provide a link to this publication

The links have been provided for these publications

Minor revisions (We can make these changes on resubmission, although it will speed up publication of your manuscript if you do them while making the major changes above).

MS ID - Please remove this header

Change has been made
Title - Please remove this text from before the Title
Change has been made
Institution where work carried out - Please move this to the Acknowledgements
Change has been made
Please ensure that abstract in our submission system matches exactly the
abstract in your manuscript
Done
Materials and Methods - Please rename this section `Methods'
Change has been made
Tables - Please move these to a position after the references
Change has been made
Page 20 is blank - Please remove this
Change has been made
Please take this opportunity to check your manuscript for any typographical
errors and to make any final corrections or revisions as this is the final proofing
stage for your manuscript, and you will not be able to make changes after
acceptance.
Manuscript checked for errors, thanks!

Many thanks again for your assistance and professional service you provide to
authors.

Prof. MF Chersich